Author's response to reviews

Title: Outcome measures of the 6 minute walk test: relationships with physiologic and computed tomography findings in patients with sarcoidosis

Authors:

Esam H Alhamad (esamalhamad@yahoo.com)
Shaffi Ahmad Shaik (shaffi786@yahoo.com)
Majdy M Idrees (majiidrees@gmail.com)
Mohammed O Alanezi (alanezi@hotmail.com)
Arthur C Isnani (acisnanimd@yahoo.com)

Version: 11 Date: 9 August 2010

Author's response to reviews:

Please note I have amended the legends for Table 4 and Table 5 as indicated below, and minor typographical errors were detected and corrected in Table 5 and Table 6 legends.

Table 4. Simple regression analysis: Relationship between distance and quantitative study variables

Table 5. Simple regression analysis: Relationship between DSP and quantitative variables

Sincerely,